Author's response to reviews

Title: Midwifery tutor capacity and willingness to teach contraception, post-abortion care, and legal pregnancy termination in Ghana

Authors:

Gertrude Voetagbe (gobanahene@yahoo.co.uk)
Nathaniel Yellu (Nathaniel.Yellu@hru-ghs.org)
Joseph Mills (millsi@ipas.org)
Ellen Mitchell (ellenmaevemitchell@gmail.com)
Amanda Adu-Amankwah (amankwaha@ipas.org)
Felix Nyante (kwamenyante@yahoo.com)
Koma Jehu-Appiah (jehuk@ipas.org)

Version: 8 Date: 18 February 2010

Author's response to reviews:

The managing Editor
Human Resources for Health Journal
World Health Organisation

Dear Sir/madam

Manuscript: Filling the Gaps in Ghana: Midwifery Tutor Capacity and Willingness to Teach Contraception, Post Abortion Care and Legal Pregnancy Termination

Thank you for giving us the chance for proof reading the manuscript. All typographical issues and final corrections on the content have made.